Author's response to reviews

Title: Guidelines for the use and management of long-acting injectable antipsychotics in serious mental illness

Authors:

Pierre Michel Llorca (pmllorca@chu-clermontferrand.fr)
Mocrane Abbar (mocrane.ABBAR@chu-nimes.fr)
Philippe Courtet (philippe.courtet@univ-montp1.fr)
Sebastien Guillaume (s-guillaume@chu-montpellier.fr)
Sylvie Lancrenon (s.lancrenon@sylia-stat.fr)
Ludovic Samalin (lsamalin@chu-clermontferrand.fr)

Version: 3 Date: 4 December 2013

Author's response to reviews: see over
To: Chief Editor

BMC Psychiatry

Sir,

Please, find enclosed a second revised version of manuscript entitled "Guidelines for the use and management of long-acting injectable antipsychotics in serious mental illness" for publication in BMC Psychiatry.

Hoping to hear from you soon, I remain

Sincerely Yours

Ludovic Samalin
The comments have been addressed adequately. Before final acceptance the authors should correct the term “schizophrenic patient” to “patient with schizophrenia”.

RESPONSE: We have corrected the term “schizophrenic patient” to “patient with schizophrenia. The changes are highlighted in the text of the manuscript.